Author's response to reviews

Title: Catheter-related bacteremia due to Kocuria rosea in a patient undergoing peripheral blood stem cell transplantation

Authors:

Fevzi Altuntas (faltuntas@hotmail.com)
Orhan Yildiz (oyildiz@erciyes.edu.tr)
Bulent Eser (beser@erciyes.edu.tr)
Kursat Gundogan (kursatgundogan@hotmail.com)
Bulent Sumerkan (sumerkan@erciyes.edu.tr)
Mustafa Cetin (mcetin@erciyes.edu.tr)

Version: 3 Date: 11 December 2004

Author's response to reviews:

Dear Editor,
I am pleased to be accepted for publication last version of our article entitled "Catheter-related bacteremia due to Kocuria rosea in a patient undergoing peripheral blood stem cell transplantation". We checked the text and clearly answered the editorial queries:
1. Running title "Bacteremia due to Kocuria rosea in Stem Cell Transplantation" is removed from the title page.
2. Each affiliation is written in full, regardless of whether some authors are associated with different departments within the same institution.
3. E-mail addresses for all authors are included on the title page.
4. Key words section "bacteremia, catheter, stem cell transplantation, Kocuria rosea" is removed from the manuscript.
5. The authors declare that they have no competing interests
I very much look forward to hearing from you.
With warmest regards.
Orhan Yildiz, MD.